Author's response to reviews

Title: Effect of the kangaroo position on the electromyographic activity of preterm children: a cohort study.

Authors: 
kaísa Trovão Diniz (kaisa.tdz@hotmail.com)
José Eulálio Cabral Filho (eulalio@imip.org.br)
Rafael Moura Miranda (rafael.mm@hotmail.com)
Geisy Maria Souza Lima (canguru@imip.org.br)
Danilo de Almeida Vasconcelos (osteopatia@gmail.com)

Version: 3 Date: 2 April 2013

Author's response to reviews:

Sorry, we realized they were missing important minor changes. Thus, we are sending a new version of the manuscript.

Discretionary Revisions


2. In the abstract, methods section, prior to biceps brachii you should place “the”; remove the word immediately when referring to 24h, 48h, 72h, 96h

3. Manuscript
   a. Background
      i. First paragraph, last sentence, “The skin to skin contact and the vertical position the child adopts during this method may…
      ii. Second paragraphs, second sentence, exhibit muscular hypotonia with posturing that is manifested mostly in extension, rather than in flexion, that…